Author's response to reviews

Title: Tropical pyomyositis as a presenting feature of sub-clinical leukaemia, a case report

Authors:

Mitrakrishnan R Navinan (rayno.navinan@gmail.com)
Jevon Yudhisdran (yudhishdran@yahoo.com)
Thambyaiah Kandeepan (kathambyaiah@yahoo.com)
Aruna Kulatunga (kulatunga.aruna@yahoo.com)

Version: 2 Date: 30 July 2014

Author's response to reviews:

The editor,
Journal of medical case reports

Dear sir/Madam

I apologise, I have made the relevant changes ensuring the manuscript keeps to journal standards.

1. CARE checklist completed and attached
2. Man was changed to male
3. Background was changed to introduction
4. Keywords included in the appropriate section

Thank you

sincerely

Rayno